Author's response to reviews

Title: Provider performance in treating poor patients - Factors influencing prescribing practices in Lao PDR: a cross-sectional study

Authors:

Lamphone Syhakhang (syhakhangl@yahoo.com)
Douangdao Soukaloun (dsoukaloun@yahoo.com)
Göran Tomson (goran.tomson@ki.se)
Max Petzold (max.petzold@nhv.se)
Clas Rehnberg (clas.rehnberg@ki.se)
Rolf Wahlström (rolf.wahlstrom@ki.se)

Version: 4 Date: 24 December 2010

Author's response to reviews: see over
2010-12-23

To
The Editor
BMC Health Service Research

Revised manuscript

MS:  1469649334180061
Provider performance in treating poor patients - Factors influencing prescribing practices in Lao PDR: a cross-sectional study
Lamphone Syhakhang, Douangdao Soukaloun, Göran Tomson, Max Petzold, Clas Rehnberg and Rolf Wahlström

Dear, the final adjustments of the papers have now been done:

- The author list is now in the preferred style “First name Last name”. However, due to my problems to access the manuscript at my BMC I have not been able to check if I added any initials in the online submission system. Sorry for that.

- After the list of authors we have added the sentence “All authors have read and approved the final manuscript.”

- We have removed the visible vertical line from table 2.

We also went back to the formatting checklist once more.

Sincerely, Max Petzold

Nordic School of Public Health